Author’s response to reviews

Title: Mistaken perception of lipid intake and its effects: a randomized trial

Authors:

Mariana Menezes (marysnut@gmail.com)
Sueli Mingoti (suelimngt@gmail.com)
Raquel Mendonça (raqueldmendonca@gmail.com)
Aline Lopes (alinelopesenf@gmail.com)

Version: 3 Date: 02 Feb 2017

Author’s response to reviews:

University of Minas Gerais
Department of Nutrition
Journal: BMC Nutrition
Manuscript Title: Mistaken perception of lipid intake and its effects: a randomized trial
Manuscript number: NUTN-D-16-00029

Reviewer report:

Thank you for submitting the revised version of your manuscript "Mistaken perception of lipid intake and its effects: a randomized trial" (NUTN-D-16-00029R2). It has been re-assessed by an Editorial Board member and I am pleased to inform you that your manuscript has been provisionally recommended for acceptance in BMC Musculoskeletal Disorders. However, before we can pass your manuscript on to our production team, there are a couple of minor stylistic points that need to be addressed as outlined below:

Thank you. We think just happened a typing error in the text above: the manuscript was submitted and has been revised by BMC Nutrition, not BMC Musculoskeletal Disorders.

1) The results are based on the findings from a randomized controlled trial, but no indication is given in the manuscript about whether the trial has been registered in a publicly available
registry. Please provide the trial registration number and date of registration in the first part of the Methods section (lines 100-104)

When the research project was beginning its execution (2008), the registration of trials in publicly available registry was still not a standard practice in Brazil. The registration of clinical research began to be regulated only in 2012 (http://www.ensaiosclinicos.gov.br/about/). Nevertheless, it is important to highlight that the trial followed all the scientific recommendations (as Consort Statement), presenting a robust methodology. Here are some publications regarding the study:


2) Authors contributions: please use author initials rather than full names, i.e. MCM, SAM etc.

Fixed. Thanks.

3) Additional files: please move the description of the file out of the Declarations section so that it appear after line 539 on page 25

Corrected.

4) The editing certificate does not need to be uploaded again.

Noted, I could not remove it from your system, but I did not included it in the revision, so I believe it was deleted.

With kind regards,

Mariana Carvalho de Menezes, on behalf of all co-authors.